Acceleration of Deep Subsurface Fluid Fluxes in the Anthropocene

Grant Ferguson¹, Lydia Bailey², Ji-Hyun Kim³, Magdalena Osburn⁴, Peter W Reiners², Henrik Drake⁵, Bradley Stevenson⁴, and Jennifer McIntosh²

¹University of Saskatchewan ²University of Arizona ³University of Calgary ⁴Northwestern University ⁵Linnaeus University

March 25, 2024

Acceleration of Deep Subsurface Fluid Fluxes in the Anthropocene

Grant Ferguson^{1,2}, Lydia R. Bailey³, Ji-Hyun Kim⁴, Magdalena R. Osburn⁵, Peter W. Reiners⁶, Henrik Drake⁷, Bradley S. Stevenson⁵ and Jennifer C. McIntosh^{1,2}

- Civil, Geological and Environmental Engineering, University of Saskatchewan, Saskatoon, Canada
- 2. Hydrology and Atmospheric Sciences, University of Arizona, Tucson, USA
- 3. Harvard University, Department of Earth & Planetary Sciences, Cambridge, USA
- 4. Korea Institute of Geoscience and Mineral Resources, Daejeon, South Korea
- 5. Earth and Planetary Sciences, Northwestern University, Chicago, USA
- 6. Geosciences, University of Arizona, Tucson, USA
- 7. Department of Biology and Environmental Science, Linnaeus University, Kalmar, Sweden

Abstract

The Anthropocene has been framed around humanity's impact on atmospheric, biologic, and near-surface processes, such as land use and vegetation change, greenhouse gas emissions, and the above-ground hydrologic cycle. Groundwater extraction has lowered water tables in many key aquifers but comparatively little attention has been given to the impacts in the deeper subsurface. Here, we show that fluid fluxes from the extraction and injection of fluids associated with oil and gas production and inflow of water into mines likely exceed background flow rates in deep (>500 m) groundwater systems at a global scale. Projected carbon capture and sequestration (CCS), geothermal energy production, and lithium extraction to facilitate the energy transition will require fluid production rates exceeding current oil and co-produced water extraction. Natural analogs and geochemical modeling indicate that subsurface fluid manipulation in the Anthropocene will likely appear in the rock record. The magnitude and importance of these changes are unclear, due to a lack of understanding of how deep subsurface hydrologic and geochemical cycles and associated microbial life interact with the rest of the Earth system.

Key Points

- Current anthropogenic fluid fluxes in the deep subsurface likely exceed background fluxes.
- Anthropogenic fluid fluxes in the deep subsurface are expected to accelerate with the energy transition.
- Injection and production of fluids from the deep subsurface is expected to leave a mark on the geologic record.

Plain Language Summary

The Anthropocene is often framed in terms of changes in climate, ecosystems and land use. These have been accompanied by changes in the Earth's water cycle, including depleted groundwater storage due to pumping in many regions. The scale of anthropogenic change in the subsurface at depths beyond typical water wells has received less attention. Fluid flow rates associated with oil and gas production likely exceed natural groundwater flow rates at depths greater than 500 m. Anthropogenic impacts to this deeper zone of the Earth's subsurface are expected to increase dramatically as we look to store carbon, mine lithium from deep brines and produce geothermal energy as part of the ongoing energy transition.

Introduction

The Anthropocene is often thought of in terms of land use change, greenhouse gas emissions and climate change, biodiversity and the appearance of distinctive physical and chemical features in the stratigraphic record (Crutzen, 2002; Lewis & Maslin, 2015; McCarthy et al., 2023; Seddon et al., 2016). The atmosphere has changed dramatically since the Industrial Revolution with rising carbon dioxide and methane concentrations (Crutzen, 2002). Land use change has resulted in substantial increases in erosion (Borrelli et al., 2017). Excavations and boreholes are widespread (Zalasiewicz et al., 2014), particularly in urban environments (Melo Zurita et al., 2018). Combined with aggregate extraction for building materials, humans are the largest geomorphologic agent on Earth (Syvitski et al., 2022). The hydrologic cycle has also been profoundly altered at a global scale, with changes in soil moisture, surface water, the cryosphere and groundwater at scales impacting the Earth system (Gleeson et al., 2020). How the Anthropocene is manifested in the deeper subsurface, below typical depths of current groundwater extraction (>~500 m), has received less attention (Melo Zurita et al., 2018). Pores and fractures at these depths contain the largest volume of water aside from the ocean (Ferguson et al., 2021) and may contain ~15% of the Earth's biomass (Bar-On et al., 2018). Groundwater residence times exceeding one million years have been found in a variety of geological settings (Ferguson et al., 2023; Warr et al., 2018), indicating that these deep subsurface ecosystems have been isolated for prolonged periods of geologic time in this "hidden" part of the Earth system that has minimal interaction with the rest of the hydrologic cycle (Warr et al., 2018). Continental to global scale studies tend to treat the subsurface as a black box that is capable of storing or producing fluids without considering how fluxes and microbial communities might change within the subsurface.

Anthropogenic impacts in the deeper subsurface are and will likely continue to be dominated by the production and injection of fluids; extraction of groundwater (Konikow, 2011; Rodell et al., 2018) and oil and gas (BP, 2022; C. Clark & Veil, 2009; McIntosh & Ferguson, 2019) already account for a substantial fraction of deep subsurface fluid fluxes. Subsurface fluid extraction and injection will accelerate with rapidly growing production of lithium (Kumar et al., 2019), helium (Cao et al., 2022), geothermal energy (Nardini, 2022), and storage and production of hydrogen (Miocic et al., 2023) and compressed air (Olabi et al., 2021), as well as, and likely most important, carbon capture and sequestration (CCS) (Benson & Cole, 2008; Krevor et al., 2023; Zoback & Smit, 2023) (Figure 1). Here, we evaluate how fluid fluxes in the Earth's deep subsurface have been affected to date, along with how they are expected to change over the coming century and how this might affect geochemical cycles and microbial communities.

Current Uses of the Subsurface

Groundwater systems have been profoundly affected during the Anthropocene. Approximately 1,000 km³/yr of groundwater is extracted each year (Wada et al., 2010). While this volume is only ~5 to 17% of global groundwater recharge, where fluxes of 6,000 to 20,000 km³/yr have been estimated (Döll & Fiedler, 2007; Gleeson et al., 2016; Wada et al., 2010), it has resulted in widespread and substantial losses of groundwater storage, which can now be tracked at monthly scales with remote sensing such as the GRACE satellite project (Rodell et al., 2018). Approximately 3,500 km³ of groundwater depletion occurred globally between 1900 and 2008 (Konikow, 2011). The extracted groundwater in excess of depletion has largely been balanced by loss of streamflow (Konikow & Leake, 2014). Most extracted groundwater is from

wells less than ~35 m deep (Jasechko & Perrone, 2021). Pumping appears to be causing an acceleration of the shallow subsurface hydrologic cycle through increases in hydraulic gradients, as modern water (i.e. containing ³H from nuclear weapons testing (Gleeson et al., 2016)) is reaching greater depths in areas where large volumes of groundwater have been extracted (Thaw et al., 2022). The corollary of this is that groundwaters that were recharged several millennia ago or longer (GebreEgziabher et al., 2022) are being reconnected with the rest of the hydrologic cycle.

The deeper subsurface (defined here as >500 m) has been more profoundly affected than shallower realms when background conditions are compared to anthropogenic activities. Fluid volumes deeper than 500 m likely exceed 30 million km³ (Ferguson et al., 2021) but these fluids are weakly connected to the rest of the hydrologic cycle under natural conditions, with estimated fluxes of <13 km³/yr (Ferguson et al., 2023) (Figure 2). Between 1970 and 2020, approximately 200 km³ of oil was produced globally (IEA, 2021b). For every 1 m³ of oil extracted from the subsurface, approximately 3-5 m³ of water is co-produced (C. Clark & Veil, 2009), resulting in a total fluid volume of 1,000 km³. The approximately 20 km³/yr of fluid produced by the oil industry during that 50 year time period likely exceeds any background fluid fluxes at depths between 500 m and a few km in sedimentary basins. Overall fluid budgets in these environments are often near zero because the co-produced water and additional water for reservoir pressure maintenance (i.e. waterflooding) or hydraulic fracturing is injected into the subsurface. However, at subregional scales the production and injection of fluids often results in large changes in hydraulic gradients (Jellicoe et al., 2022).

Environmental concerns surrounding fluids in the deep subsurface have focused on upward leakage into the rest of the hydrologic cycle and the atmosphere (Dusseault & Jackson, 2014; Kang et al., 2014; Lacombe et al., 1995; Perra et al., 2022). However, impacts to the deep subsurface itself will also occur because the chemical and microbial composition of injected fluids differ from *in situ* fluids. Water injected for hydraulic fracturing and secondary recovery (waterflooding) is often seawater, surface water, or shallow groundwater (Bayona, 1993; Kondash & Vengosh, 2015; Scanlon et al., 2019) with various additives (e.g. biocides, corrosion inhibitors) (Elsner & Hoelzer, 2016). Produced and flowback water are often injected into other strata with different original fluid chemistries and this reinjection strategy has become increasingly common in unconventional oil and gas developments. For example, flowback and produced water from the Bakken Formation are routinely injected into the shallower Dakota/Mannville Group in the Williston Basin (Jellicoe et al., 2022; Scanlon et al., 2016) and produced water from the Mississippi Lime, a play relying on dewatering to drive gas exsolution, is injected into the deeper Arbuckle Group in Oklahoma and Kansas (Murray, 2013). However, even where produced water is injected back into its same source strata, the oxidationreduction (redox) states and microbial communities within these fluids are profoundly altered from their initial conditions. There have been no comprehensive studies examining how these changes affect solute transport, fluid chemistry and microbial activity at regional scales.

Fluid injection can have notable effects on the subsurface biosphere, by introducing new microorganisms, fluids with different chemistries and redox conditions, and/or amendments that alter *in situ* microbial communities that have coevolved with fluid and host rock properties over long time periods, in some cases 10s of millions of years or more (Castro et al., 1998;

Ferguson et al., 2018). Documentation of these anthropogenic changes to the deep biosphere has rarely been done along with tracking of produced and injected fluid volumes. One well known example is that of reservoir souring, resulting from the introduction of SO₄ via fluid injection, which can stimulate sulfate reducing microbial populations, producing H₂S and reducing fuel grade (Cord-Ruwisch et al., 1987). The common mechanisms of ameliorating this "souring," such as NO₃ injection, represent intentional modulation of the subsurface biosphere at industrial scales. Another is the introduction of *Halanaerobium* in deep hydraulically fractured shale gas reservoirs, which were previously sterile or near sterile(Booker et al., 2019). In some cases, oil and gas companies have intentionally stimulated existing microbial populations to degrade hydrocarbons and produce methane by injecting amendments, such as yeast or algal extracts and nutrients (Barnhart et al., 2022; Ritter et al., 2015). Similarly, CO₂ injection for enhanced oil recovery or storage can enhance microbial methanogenesis in some settings (McIntosh et al., 2010; Tyne et al., 2021). Preliminary research on H₂ storage suggests that this may also promote microbial activity (Dopffel et al., 2021).

The inflow of groundwater into mines and pumping to prevent these inflows also represents a substantial perturbation to deep groundwater flow. There is no comprehensive global database of inflow rates but values of 1 to 1,000 L/s have been documented (Dong et al., 2021; Greene et al., 2008; Winter et al., 1983). If an inflow rate of 10 L/s is representative of the globe's 6,000 active mines (Maus et al., 2020), this would result in 1.9 km³/yr, which is similar to the current rate of global oil production. These waters are often released to surface waters as the lower permeability environment associated with many mines prevents subsurface disposal. Changes in hydrogeochemical conditions and microbial communities in the vicinity of

mines will result from downwelling of meteoric water and upwelling of older, more saline water (Figure 1).

The Future of the Subsurface

Humanity's use of the subsurface over the next century is expected to increase to address climate change and energy security. This will include production of lithium, helium, and geothermal energy, along with storage and production of hydrogen, storage of compressed air and geologic CCS. CCS is arguably the most important of these projected uses in terms of reducing greenhouse gas emissions, with many of the studies examining the capacity to sequester carbon in the subsurface focusing on estimation of the volume of porosity in sedimentary basins suited for this purpose (Benson & Cole, 2008; Krevor et al., 2023; Zoback & Smit, 2023). Additional capacity exists in mafic and ultramafic rocks(Gislason & Oelkers, 2014) but uncertainty exists around the ability to inject large volumes of fluid into these often low permeability environments (Fisher, 1998). Global capacity in sedimentary basins may exceed 60,000 Gt (Kearns et al., 2017), which far exceeds the 220 to 2500 Gt that may need to be sequestered. Comparing this amount to historical fluid production and injection and fluid fluxes provides a different perspective.

Although some of the injected CO₂ may be quickly mineralized in rock or dissolved in fluids (Benson & Cole, 2008), if sequestered as a separate phase, 2,000 Gt of CO₂ is equivalent to a volume of ~3,300 km³. This assumes a density of 600 kg/m³ is assumed for supercritial CO₂ (Zoback & Smit, 2023), which is a typical value for the temperatures and pressure found at depths greater than 800 m (Benson & Cole, 2008). This volume of fluid is an order of magnitude larger than cumulative historical global oil production. A proposed annual sequestration rate of

6 Gt/yr (~10 km³/yr) of CO₂ by 2050 would occur at a rate 50% greater than global oil production in 2022 (Zoback & Smit, 2023) and similar to the maximum estimated global flux of deep groundwater (Ferguson et al., 2023). CO₂ injection is likely to be concentrated geographically, near anthropogenic sources of CO₂ (e.g., power plants) and in areas where suitable subsurface reservoirs exist (Bachu, 2003). Experience from oil and gas production and associated co-produced water management indicates that even where fluid budgets are close to balanced, large hydraulic head changes will occur at local scales near injection wells resulting in substantial changes in regional groundwater flow systems (Barson, 1993; Jellicoe et al., 2022) and, in some cases, induced seismicity (Peterie et al., 2018). Such impacts have yet to be documented in CCS projects but regional pressure changes exceeding 0.1 MPa have been predicted based on simulations (Birkholzer et al., 2015). Leakage of CO₂ might occur through leaky wells (Gasda et al., 2014) and faults (Song & Zhang, 2013) where they exist. Caution will need to be exercised if use of the subsurface for CCS becomes more extensive.

Produced water from oil production and other sedimentary brines have been proposed as sources of lithium (Kumar et al., 2019; Munk et al., 2016). Lithium extraction from sedimentary basin brines will only be viable if large fluid volumes can be produced, likely from wells producing at several times the rate of a typical oil well (Marza et al., 2023). The median Li concentration in sedimentary basin brines in the USA is 5 mg/L (Blondes et al., 2016) and we assume that concentrations in similar environments around the globe are comparable. At this concentration, 20 km³ of brine would be required to produce an amount equal to global Li production of 100,000 tpy in 2022 (USGS, 2023), an amount similar to current combined annual oil and associated produced water volumes (Figure 2).

Geothermal electricity production of 1,050 TW h/yr using binary technology, where a secondary working fluid is used to generate steam in lower temperature environments (>70°C and <180°C), in conventional and enhanced geothermal systems has been projected for 2050 using an integrated assessment model (van der Zwaan & Dalla Longa, 2019). Binary geothermal systems require approximately 610,000 USGPD/MW_e (= $6.53 \times 10^7 \text{ m}^3/\text{yr}/\text{MW}_e$) (C. E. Clark et al., 2010), indicating that 75 km³/yr of fluid would need to be produced to support the projected level of geothermal electricity production. This target represents a large expansion of geothermal capacity but would only account for a small fraction of current electricity generation, at 67 TWh/yr compared to the overall generation of 23,000 TWh in 2019 (IEA, 2021a).

Despite the large volume of pore space in the subsurface globally, there will inevitably be competition between different applications (Ferguson, 2013). All developments here will benefit from the presence of elevated permeability and porosity to allow for larger injection and/or extraction rates. In some cases, such as geothermal power production and CCS, overlapping temperature ranges may allow for synergistic developments (Randolph & Saar, 2011). Proposed and operational CCS projects at depths > 3,000 m in Canada (Worth et al., 2014) and China (Tang et al., 2014) have temperatures >70°C required to produce electricity with a binary geothermal system (van der Zwaan & Dalla Longa, 2019). In other cases, previous developments may complicate other types of subsequent uses. For example, strata that have previously been extensively developed for oil and gas may not be appropriate for CCS or H₂ storage because of the possibility of leakage through older wells (Gasda et al., 2014) or newer wells with construction issues (Bexte et al., 2008). Reservoirs that have a history of injection of

fluids that have spent time at the surface are likely to have cooled, which will have altered their potential to produce geothermal power or sequester carbon (Ferguson & Ufondu, 2017). The lack of characterization of impacts of fluid production and injection will be a challenge as we look to repurpose portions of the subsurface that have been previously developed. Whether this competition restricts development or expands that volume of subsurface use is unclear.

Similar magnitudes of changes to subsurface fluid budgets and associated changes in hydraulic gradients due to extraction of groundwater and hydrocarbons and injection of various fluids for storage and disposal are occurring orders of magnitude more rapidly than geological drivers. For example, groundwater flow in the Mannville Group of the central portion of the Williston Basin, Canada appeared to have been stable for millions of years, even through multiple glacial cycles (Cheng et al., 2021), yet operation of injection wells since the 1960s for disposal of oilfield produced waters has resulted in substantial disruption of background groundwater flow patterns (Jellicoe et al., 2022). The implications of these changes to groundwater flow on solute transport and microbial activity will likely occur with substantial time lags and may persist well into the future even once the anthropogenic perturbation ceases due to the long-time scales associated with hydraulic diffusion (Bredehoeft & Durbin, 2009). Responses of shallow groundwater systems to new boundary conditions associated with climate change will likely take decades to centuries (Cuthbert et al., 2019). Fluids in the deeper subsurface are slow to respond to shifts in climate and topography, with regional aquifer systems typically having hydraulic response times of thousands to millions of years (Rousseau-Gueutin et al., 2013). Solute transport responses typically take place over longer time periods due differences between rates of advection and hydraulic diffusion(Ferguson et al., 2023).

Evidence for increases in subsurface paleofluid fluxes, solute transport and microbial activity have been tied to geological events such as continental scale glaciations (McIntosh et al., 2012) or extensive denudation and incision by large rivers (Kim et al., 2022; Li et al., 2023).

There has been considerable debate about how the Anthropocene will appear in the geologic record but this has largely focused on depositional processes and what markers will delineate the shift from the Holocene to Anthropocene (McCarthy et al., 2023; Zalasiewicz et al., 2011). Anthropogenic activities in the deep subsurface will also leave a mark in the geologic record. Wells and boreholes will likely be rarely encountered due to their small diameter and large spacing (Gasda et al., 2014; Nicot, 2009; Zalasiewicz et al., 2014). This is likely to be the case even in areas of extremely high well densities, such as historical oil fields in Pennsylvania, USA (Dilmore et al., 2015) or some areas of heavy oil production in Alberta, Canada (Gasda et al., 2014), where well densities exceed 50 wells/km² locally. Hydraulic fractures, which commonly extend 50 to 100 m from the wellbore (Davies et al., 2012), will increase the footprint of human activities slightly but activities associated with more permeable strata, where fluids can migrate greater distances, are likely to leave more extensive evidence. Transport of fluids from injection into conventional oil and gas reservoirs commonly reaches several 100 m (Craig Jr et al., 1955; Wassmuth et al., 2009) and transport of CO₂ of distances of several 100 m have been observed in CCS projects (Ringrose, 2018). Contaminant plumes with greater extents can develop in shallower groundwater systems under background hydraulic gradients(Van der Kamp et al., 1994) but will be less common in deeper systems due to the lower hydraulic gradients unless injection or pumping wells are operated for long time periods (Jellicoe et al., 2022). Transport can be further enhanced by the presence of leaky wells.

Instances are documented where migration of fluids over distances of several 100 m have occurred through leaky wells in waterflooding (Eger & Vargo, 1989) and hydraulic fracturing operations(DiGiulio & Jackson, 2016). Contaminant plumes in shallow groundwater systems can persist for decades or longer (Essaid et al., 2011) and timescales in the deep subsurface could be even longer due to the smaller geochemical fluxes available to support geochemical transformations.

Secondary minerals, such as barite, carbonates and sulfides are commonly precipitated following the injection of water for secondary recovery of oil (i.e. waterflooding), hydraulic fracturing or for disposal of produced water from oil and gas operations (Bennion et al., 1998; Engle & Rowan, 2014; Jew et al., 2017). CCS operations are predicted to result in bleaching of sandstones and release of trace metals due to removal of hematite (Bickle & Kampman, 2013), along with precipitation of halite (Muller et al., 2009). At a smaller scale, calcite and sulfide precipitation may occur due to stimulation of microbial activity by materials introduced during drilling and well construction (Pidchenko et al., 2023). The isotopic signatures of these minerals precipitated due to injection of fluids may differ from similar minerals precipitated under background conditions (Śliwiński et al., 2017). The rock record in environments that have experienced fluid flow events that resulted in precipitation of secondary minerals driven by changes in solute fluxes, salinity, redox conditions and microbial communities can provide some insights into how anthropogenic activities in the deep subsurface are being preserved.

Conclusions

Extraction of groundwater as well as production and injection of fluids by the oil and gas industry have become important components of the global subsurface fluid budgets during the

Anthropocene. Increased use of subsurface fluids for extraction of energy and mineral resources and associated pore space for storage of alternative energy and anthropogenic waste has been proposed to confront climate change. While there is likely adequate subsurface storage, the fluxes of fluids involved with CCS, geothermal energy production and lithium extraction will be substantial, likely exceeding current levels associated with the oil and gas industry.

The subsurface and its pore space has often been viewed as a resource (Melo Zurita et al., 2018) rather than part of the Earth system. Over the past two decades, there has been an increase in the awareness of the microbial communities that inhabit the deep subsurface of depths of up to a few km (Bar-On et al., 2018; Magnabosco et al., 2018; McMahon & Parnell, 2014). This has been accompanied by questions about how the deep subsurface fits within the larger Earth system in terms of microbial life and associated water and geochemical fluxes (Ferguson et al., 2021, 2023; Lollar et al., 2019; Warr et al., 2018). As we stand at the precipice of the energy transition, we have the opportunity to develop the deep subsurface in a manner that allows us to study its natural functions and response to anthropogenic perturbations to minimize human impacts and build understanding, synergies and resilience.

Data Availability Statement

No new data was generated or compiled to support writing of this commentary.

Acknowledgments

The authors are grateful to an anonymous reviewer who provided comments that helped to improve this manuscript. This research was supported by an NSERC Discovery Grant to Ferguson and an NSF FRES grant (2120733 McIntosh, Reiners, Drake; 2120912 Osburn).

References

- Bachu, S. (2003). Screening and ranking of sedimentary basins for sequestration of CO2 in geological media in response to climate change. *Environmental Geology*, 44(3), 277–289.
- Barnhart, E. P., Ruppert, L. F., Hiebert, R., Smith, H. J., Schweitzer, H. D., Clark, A. C., et al.
 (2022). In situ enhancement and isotopic labeling of biogenic coalbed methane. *Environmental Science & Technology*, *56*(5), 3225–3233.
- Bar-On, Y. M., Phillips, R., & Milo, R. (2018). The biomass distribution on Earth. *Proceedings of the National Academy of Sciences*, *115*(25), 6506–6511.
- Barson, D. (1993). *The hydrogeological characterization of oil fields in north-central Alberta for exploration purposes* (Ph.D. Thesis). University of Alberta, Edmonton, Canada. Retrieved from University of Alberta Education and Research Archive.
- Bayona, H. J. (1993). A review of well injectivity performance in Saudi Arabia's Ghawar field seawater injection program. In *SPE Middle East Oil and Gas Show and Conference* (p. SPE-25531-MS). SPE.
- Bennion, D. B., Bennion, D. W., Thomas, F. B., & Bietz, R. F. (1998). Injection water quality a key factor to successful waterflooding. *Journal of Canadian Petroleum Technology*, *37*(06).
- Benson, S. M., & Cole, D. R. (2008). CO₂ sequestration in deep sedimentary formations. *Elements*, *4*(5), 325–331.
- Bexte, D. C., Willis, M., De Bruijn, G. G., Eitzen, B., & Fouillard, E. (2008). Improved cementing practice prevents gas migration. *World Oil Magazine*, *229*(6), 73–75.

- Bickle, M., & Kampman, N. (2013). Lessons in carbon storage from geological analogues. *Geology*, 41(4), 525–526.
- Birkholzer, J. T., Oldenburg, C. M., & Zhou, Q. (2015). CO2 migration and pressure evolution in deep saline aquifers. *International Journal of Greenhouse Gas Control*, 40, 203–220.
- Blondes, M. S., Gans, K. D., Thordsen, J. J., Reidy, M. E., Thomas, B., Engle, M. A., et al. (2016).
 US Geological Survey National Produced Waters Geochemical Database v2. 3
 (PROVISIONAL). United States Geological Survey.
- Booker, A. E., Hoyt, D. W., Meulia, T., Eder, E., Nicora, C. D., Purvine, S. O., et al. (2019). Deepsubsurface pressure stimulates metabolic plasticity in shale-colonizing Halanaerobium spp. *Applied and Environmental Microbiology*, *85*(12), e00018-19.
- Borrelli, P., Robinson, D. A., Fleischer, L. R., Lugato, E., Ballabio, C., Alewell, C., et al. (2017). An assessment of the global impact of 21st century land use change on soil erosion. *Nature Communications*, *8*(1), 2013.
- BP. (2022). BP Statistical Review of World Energy 2021. British Petroleum Statistical Review of World Energy, Bplc. Editor, Pureprint Group Limited, UK.
- Bredehoeft, J. D., & Durbin, T. (2009). Ground water development—The time to full capture problem. *Groundwater*, *47*(4), 506–514.
- Cao, Q., Li, Y., Fang, C., Liu, R., Xiao, H., & Wang, S. (2022). Status quo and utilization trend of global helium resources. *Frontiers in Environmental Science*, *10*, 1028471.
- Castro, M. C., Jambon, A., De Marsily, G., & Schlosser, P. (1998). Noble gases as natural tracers of water circulation in the Paris Basin: 1. Measurements and discussion of their origin

and mechanisms of vertical transport in the basin. *Water Resources Research*, 34(10), 2443–2466.

- Cheng, A., Lollar, B. S., Warr, O., Ferguson, G., Idiz, E., Mundle, S. O., et al. (2021). Determining the role of diffusion and basement flux in controlling 4He distribution in sedimentary basin fluids. *Earth and Planetary Science Letters*, *574*, 117175.
- Clark, C., & Veil, J. (2009). *Produced water volumes and management practices in the United States*. Argonne National Laboratory (ANL).
- Clark, C. E., Harto, C. B., Sullivan, J. L., & Wang, M. Q. (2010). *Water use in the development and operation of geothermal power plants*. Argonne National Lab.(ANL), Argonne, IL (United States).
- Cord-Ruwisch, R., Kleinitz, W., & Widdel, F. (1987). Sulfate-reducing bacteria and their activities in oil production. *Journal of Petroleum Technology*, *39*(01), 97–106.
- Craig Jr, F. F., Geffen, T. M., & Morse, R. A. (1955). Oil recovery performance of pattern gas or water injection operations from model tests. *Transactions of the AIME*, 204(01), 7–15.

Crutzen, P. J. (2002). Geology of mankind. Nature, 415(6867), 23–23.

https://doi.org/10.1038/415023a

- Cuthbert, M. O., Gleeson, T., Moosdorf, N., Befus, K. M., Schneider, A., Hartmann, J., & Lehner, B. (2019). Global patterns and dynamics of climate–groundwater interactions. *Nature Climate Change*, *9*(2), 137–141.
- Davies, R. J., Mathias, S. A., Moss, J., Hustoft, S., & Newport, L. (2012). Hydraulic fractures: How far can they go? *Marine and Petroleum Geology*, *37*(1), 1–6.

- DiGiulio, D. C., & Jackson, R. B. (2016). Impact to Underground Sources of Drinking Water and Domestic Wells from Production Well Stimulation and Completion Practices in the Pavillion, Wyoming, Field. *Environmental Science & Technology*, *50*(8), 4524–4536. https://doi.org/10.1021/acs.est.5b04970
- Dilmore, R. M., Sams III, J. I., Glosser, D., Carter, K. M., & Bain, D. J. (2015). Spatial and temporal characteristics of historical oil and gas wells in Pennsylvania: Implications for new shale gas resources. *Environmental Science & Technology*, *49*(20), 12015–12023.
- Döll, P., & Fiedler, K. (2007). Global-scale modeling of groundwater recharge. *Hydrology and Earth System Sciences Discussions*, *4*(6), 4069–4124.
- Dong, S., Wang, H., Guo, X., & Zhou, Z. (2021). Characteristics of water hazards in China's coal mines: a review. *Mine Water and the Environment*, *40*, 325–333.
- Dopffel, N., Jansen, S., & Gerritse, J. (2021). Microbial side effects of underground hydrogen storage–Knowledge gaps, risks and opportunities for successful implementation. *International Journal of Hydrogen Energy*, *46*(12), 8594–8606.
- Dusseault, M., & Jackson, R. (2014). Seepage pathway assessment for natural gas to shallow groundwater during well stimulation, in production, and after abandonment. *Environmental Geosciences*, *21*(3), 107–126.
- Eger, C. K., & Vargo, J. S. (1989). Prevention: Ground Water Contamination at the Martha Oil Field, Lawrence and Johnson Counties, Kentucky. In *Environmental Concerns in the Petroleum Industry*. AAPG.

- Elsner, M., & Hoelzer, K. (2016). Quantitative survey and structural classification of hydraulic fracturing chemicals reported in unconventional gas production. *Environmental Science* & *Technology*, *50*(7), 3290–3314.
- Engle, M. A., & Rowan, E. L. (2014). Geochemical evolution of produced waters from hydraulic fracturing of the Marcellus Shale, northern Appalachian Basin: A multivariate compositional data analysis approach. *International Journal of Coal Geology*, *126*, 45–56.
- Essaid, H. I., Bekins, B. A., Herkelrath, W. N., & Delin, G. N. (2011). Crude oil at the Bemidji site: 25 years of monitoring, modeling, and understanding. *Groundwater*, *49*(5), 706–726.
- Ferguson, G. (2013). Subsurface energy footprints. *Environmental Research Letters*, 8(1). https://doi.org/10.1088/1748-9326/8/1/014037
- Ferguson, G., & Ufondu, L. (2017). Geothermal energy potential of the Western Canada Sedimentary Basin: Clues from coproduced and injected water. *Environmental Geosciences*, 24(3), 113–121.
- Ferguson, G., McIntosh, J. C., Grasby, S. E., Hendry, M. J., Jasechko, S., Lindsay, M. B. J., & Luijendijk, E. (2018). The Persistence of Brines in Sedimentary Basins. *Geophysical Research Letters*, 45(10), 4851–4858.
- Ferguson, G., McIntosh, J. C., Warr, O., Sherwood Lollar, B., Ballantine, C. J., Famiglietti, J. S., et al. (2021). Crustal Groundwater Volumes Greater than Previously Thought. *Geophysical Research Letters*, e2021GL093549.

- Ferguson, G., McIntosh, J. C., Jasechko, S., Kim, J.-H., Famiglietti, J. S., & McDonnell, J. J. (2023). Groundwater deeper than 500 m contributes less than 0.1% of global river discharge. *Communications Earth & Environment*, *4*(1), 48.
- Fisher, A. T. (1998). Permeability within basaltic oceanic crust. *Reviews of Geophysics*, *36*(2), 143–182.
- Gasda, S. E., Bachu, S., & Celia, M. A. (2014). Spatial characterization of the location of potentially leaky wells penetrating a deep saline aquifer in a mature sedimentary basin. *Environmental Geology*. https://doi.org/10.1007/s00254-004-1073-5
- GebreEgziabher, M., Jasechko, S., & Perrone, D. (2022). Widespread and increased drilling of wells into fossil aquifers in the USA. *Nature Communications*, *13*(1), 1–12.
- Gislason, S. R., & Oelkers, E. H. (2014). Carbon storage in basalt. *Science*, 344(6182), 373–374.
- Gleeson, T., Befus, K. M., Jasechko, S., Luijendijk, E., & Cardenas, M. B. (2016). The global volume and distribution of modern groundwater. *Nature Geoscience*, *9*(2), 161–167.
- Gleeson, T., Wang-Erlandsson, L., Porkka, M., Zipper, S. C., Jaramillo, F., Gerten, D., et al. (2020). Illuminating water cycle modifications and Earth system resilience in the Anthropocene. *Water Resources Research*, *56*(4), e2019WR024957.
- Greene, S., Battye, N., Clark, I., Kotzer, T., & Bottomley, D. (2008). Canadian Shield brine from the Con Mine, Yellowknife, NT, Canada: Noble gas evidence for an evaporated
 Palaeozoic seawater origin mixed with glacial meltwater and Holocene recharge. *Geochimica et Cosmochimica Acta*, 72(16), 4008–4019.
- IEA. (2021a). *Electricity Information: Overview*. Paris. Retrieved from https://www.iea.org/reports/electricity-information-overview

- IEA. (2021b). World crude oil production by region, 1971-2020. Paris. Retrieved from https://www.iea.org/data-and-statistics/charts/world-crude-oil-production-by-region-1971-2020
- Jasechko, S., & Perrone, D. (2021). Global groundwater wells at risk of running dry. *Science*, *372*(6540), 418–421.
- Jasechko, S., Perrone, D., Befus, K. M., Cardenas, M. B., Ferguson, G., Gleeson, T., et al. (2017). Global aquifers dominated by fossil groundwaters but wells vulnerable to modern contamination. *Nature Geoscience*, *10*(6), 425–429.
- Jellicoe, K., McIntosh, J. C., & Ferguson, G. (2022). Changes in deep groundwater flow patterns related to oil and gas activities. *Groundwater*, *60*(1), 47–63.
- Jew, A. D., Dustin, M. K., Harrison, A. L., Joe-Wong, C. M., Thomas, D. L., Maher, K., et al. (2017). Impact of organics and carbonates on the oxidation and precipitation of iron during hydraulic fracturing of shale. *Energy & Fuels*, *31*(4), 3643–3658.
- Kang, M., Kanno, C. M., Reid, M. C., Zhang, X., Mauzerall, D. L., Celia, M. A., et al. (2014). Direct measurements of methane emissions from abandoned oil and gas wells in Pennsylvania.
 Proceedings of the National Academy of Sciences, *111*(51), 18173–18177.
- Kearns, J., Teletzke, G., Palmer, J., Thomann, H., Kheshgi, H., Chen, Y.-H. H., et al. (2017). Developing a consistent database for regional geologic CO2 storage capacity worldwide. *Energy Procedia*, *114*, 4697–4709.
- Kim, J.-H., Ferguson, G., Person, M., Jiang, W., Lu, Z.-T., Ritterbusch, F., et al. (2022). Krypton-81 dating constrains timing of deep groundwater flow activation. *Geophysical Research Letters*, e2021GL097618.

- Kondash, A., & Vengosh, A. (2015). Water footprint of hydraulic fracturing. *Environmental* Science & Technology Letters, 2(10), 276–280.
- Konikow, L. F. (2011). Contribution of global groundwater depletion since 1900 to sea-level rise. *Geophysical Research Letters*, *38*(17).
- Konikow, L. F., & Leake, S. A. (2014). Depletion and Capture: Revisiting "The Source of Water Derived from Wells." *Groundwater*, *52*(S1), 100–111.
- Krevor, S., de Coninck, H., Gasda, S. E., Ghaleigh, N. S., de Gooyert, V., Hajibeygi, H., et al. (2023). Subsurface carbon dioxide and hydrogen storage for a sustainable energy future. *Nature Reviews Earth & Environment*, 1–17.
- Kumar, A., Fukuda, H., Hatton, T. A., & Lienhard, J. H. (2019). Lithium recovery from oil and gas produced water: a need for a growing energy industry. *ACS Energy Letters*, *4*(6), 1471– 1474.
- Lacombe, S., Sudicky, E., Frape, S., & Unger, A. (1995). Influence of Leaky Boreholes on Cross-Formational Groundwater Flow and Contaminant Transport. *Water Resources Research*, *31*(8), 1871–1882.
- Lemay, T. G. (2008). *Description of the Process for Defining the Base of Groundwater Protection* (p. 27). Edmonton, Alberta.

Lewis, S. L., & Maslin, M. A. (2015). Defining the Anthropocene. *Nature*, *519*(7542), 171–180.

Li, J., Pang, Z., Liu, Y., Hu, S., Jiang, W., Tian, L., et al. (2023). Changes in groundwater dynamics and geochemical evolution induced by drainage reorganization: Evidence from 81Kr and 36Cl dating of geothermal water in the Weihe Basin of China. *Earth and Planetary Science Letters*, *623*, 118425.

- Lollar, G. S., Warr, O., Telling, J., Osburn, M. R., & Sherwood Lollar, B. (2019). 'Follow the Water': Hydrogeochemical Constraints on Microbial Investigations 2.4 km Below Surface at the Kidd Creek Deep Fluid and Deep Life Observatory. *Geomicrobiology Journal*, 1–14.
- Magnabosco, C., Lin, L.-H., Dong, H., Bomberg, M., Ghiorse, W., Stan-Lotter, H., et al. (2018). The biomass and biodiversity of the continental subsurface. *Nature Geoscience*, *11*(10), 707.
- Marza, M., Ferguson, G., Thorson, J., Barton, I., Kim, J.-H., Ma, L., & McIntosh, J. (2023). Geological controls on lithium production from basinal brines across North America. *Journal of Geochemical Exploration*, 107383.
- Maus, V., Giljum, S., Gutschlhofer, J., da Silva, D. M., Probst, M., Gass, S. L., et al. (2020). A global-scale data set of mining areas. *Scientific Data*, 7(1), 289.
- McCarthy, F. M., Patterson, R. T., Head, M. J., Riddick, N. L., Cumming, B. F., Hamilton, P. B., et al. (2023). The varved succession of Crawford Lake, Milton, Ontario, Canada as a candidate Global boundary Stratotype Section and Point for the Anthropocene series. *The Anthropocene Review*, *10*(1), 146–176.
- McIntosh, J. C., & Ferguson, G. (2019). Conventional Oil—The Forgotten Part of the Water-Energy Nexus. *Groundwater*, *57*(5), 669–677.

McIntosh, J. C., & Ferguson, G. (2021). Deep Meteoric Water Circulation in Earth's Crust. *Geophysical Research Letters*, 48(5), e2020GL090461. https://doi.org/10.1029/2020GL090461

- McIntosh, J. C., Warwick, P. D., Martini, A. M., & Osborn, S. G. (2010). Coupled hydrology and biogeochemistry of Paleocene–Eocene coal beds, northern Gulf of Mexico. *GSA Bulletin*, *122*(7–8), 1248–1264.
- McIntosh, J. C., Schlegel, M., & Person, M. (2012). Glacial impacts on hydrologic processes in sedimentary basins: evidence from natural tracer studies. *Geofluids*, *12*(1), 7–21.
- McMahon, S., & Parnell, J. (2014). Weighing the deep continental biosphere. *FEMS Microbiology Ecology*, *87*(1), 113–120.
- Melo Zurita, M. de L., George Munro, P., & Houston, D. (2018). Un-earthing the subterranean Anthropocene. *Area*, *50*(3), 298–305.
- Miocic, J., Heinemann, N., Edlmann, K., Scafidi, J., Molaei, F., & Alcalde, J. (2023). Underground hydrogen storage: a review. *Geological Society, London, Special Publications*, *528*(1), SP528-2022–88.
- Muller, N., Qi, R., Mackie, E., Pruess, K., & Blunt, M. J. (2009). CO2 injection impairment due to halite precipitation. *Energy Procedia*, 1(1), 3507–3514.
- Munk, L. A., Hynek, S. A., Bradley, D., Boutt, D., Labay, K., & Jochens, H. (2016). Lithium brines: A global perspective. *Reviews in Economic Geology*, *18*, 339–365.
- Murray, K. E. (2013). State-scale perspective on water use and production associated with oil and gas operations, Oklahoma, US. *Environmental Science & Technology*, *47*(9), 4918– 4925.
- Nardini, I. (2022). Geothermal Power Generation. In *The Palgrave Handbook of International Energy Economics* (pp. 183–194). Springer International Publishing Cham.

- Nicot, J.-P. (2009). A survey of oil and gas wells in the Texas Gulf Coast, USA, and implications for geological sequestration of CO 2. *Environmental Geology*, *57*, 1625–1638.
- Olabi, A. G., Wilberforce, T., Ramadan, M., Abdelkareem, M. A., & Alami, A. H. (2021). Compressed air energy storage systems: Components and operating parameters–A review. *Journal of Energy Storage*, *34*, 102000.
- Perra, C., McIntosh, J. C., Watson, T., & Ferguson, G. (2022). Commingled Fluids in Abandoned Boreholes: Proximity Analysis of a Hidden Liability. *Groundwater*, *60*(2), 210–224.
- Peterie, S. L., Miller, R. D., Intfen, J. W., & Gonzales, J. B. (2018). Earthquakes in Kansas Induced
 by Extremely Far-Field Pressure Diffusion. *Geophysical Research Letters*, 45(3), 1395–
 1401.
- Pidchenko, I. N., Christensen, J. N., Kutzschbach, M., Ignatyev, K., Puigdomenech, I., Tullborg, E.-L., et al. (2023). Deep anoxic aquifers could act as sinks for uranium through microbialassisted mineral trapping. *Communications Earth & Environment*, *4*(1), 128.
- Randolph, J. B., & Saar, M. O. (2011). Coupling carbon dioxide sequestration with geothermal energy capture in naturally permeable, porous geologic formations: Implications for CO2 sequestration. *Energy Procedia*, *4*, 2206–2213.
- Ringrose, P. S. (2018). The CCS hub in Norway: some insights from 22 years of saline aquifer storage. *Energy Procedia*, *146*, 166–172.
- Ritter, D., Vinson, D., Barnhart, E., Akob, D. M., Fields, M. W., Cunningham, A. B., et al. (2015). Enhanced microbial coalbed methane generation: a review of research, commercial activity, and remaining challenges. *International Journal of Coal Geology*, *146*, 28–41.

- Rodell, M., Famiglietti, J. S., Wiese, D. N., Reager, J. T., Beaudoing, H. K., Landerer, F. W., & Lo, M.-H. (2018). Emerging trends in global freshwater availability. *Nature*, *557*(7707), 651– 659.
- Rousseau-Gueutin, P., Love, A. J., Vasseur, G., Robinson, N. I., Simmons, C. T., & Marsily, G. (2013). Time to reach near-steady state in large aquifers. *Water Resources Research*, *49*(10), 6893–6908. https://doi.org/10.1002/wrcr.20534
- Scanlon, B. R., Reedy, R. C., Male, F., & Hove, M. (2016). Managing the increasing water footprint of hydraulic fracturing in the Bakken Play, United States. *Environmental Science & Technology*, *50*(18), 10273–10281.
- Scanlon, B. R., Weingarten, M. B., Murray, K. E., & Reedy, R. C. (2019). Managing Basin-Scale Fluid Budgets to Reduce Injection-Induced Seismicity from the Recent U.S. Shale Oil Revolution. *Seismological Research Letters*, *90*(1), 171–182.

https://doi.org/10.1785/0220180223

- Seddon, N., Mace, G. M., Naeem, S., Tobias, J. A., Pigot, A. L., Cavanagh, R., et al. (2016).
 Biodiversity in the Anthropocene: prospects and policy. *Proceedings of the Royal Society B: Biological Sciences*, 283(1844), 20162094.
- Śliwiński, M. G., Kitajima, K., Kozdon, R., Spicuzza, M. J., Denny, A., & Valley, J. W. (2017). In situ δ13C and δ18O microanalysis by SIMS: A method for characterizing the carbonate components of natural and engineered CO2-reservoirs. *International Journal of Greenhouse Gas Control*, *57*, 116–133.
- Song, J., & Zhang, D. (2013). Comprehensive review of caprock-sealing mechanisms for geologic carbon sequestration. *Environmental Science & Technology*, *47*(1), 9–22.

- Syvitski, J., Ángel, J. R., Saito, Y., Overeem, I., Vörösmarty, C. J., Wang, H., & Olago, D. (2022). Earth's sediment cycle during the Anthropocene. *Nature Reviews Earth & Environment*, *3*(3), 179–196.
- Tang, Y., Yang, R., & Bian, X. (2014). A review of sequestration projects and application in China. *The Scientific World Journal, 2014*.
- Tester, J. W., Beckers, K. F., Hawkins, A. J., & Lukawski, M. Z. (2021). The evolving role of geothermal energy for decarbonizing the United States. *Energy & Environmental Science*, *14*(12), 6211–6241.
- Thaw, M., GebreEgziabher, M., Villafañe-Pagán, J. Y., & Jasechko, S. (2022). Modern groundwater reaches deeper depths in heavily pumped aquifer systems. *Nature Communications*, *13*(1), 5263.
- Tyne, R. L., Barry, P. H., Lawson, M., Byrne, D. J., Warr, O., Xie, H., et al. (2021). Rapid microbial methanogenesis during CO2 storage in hydrocarbon reservoirs. *Nature*, *600*(7890), 670–674.
- USGS. (2023). *Mineral commodity summaries 2023* (Report No. 2023) (p. 210). Reston, VA. https://doi.org/10.3133/mcs2023
- Van der Kamp, G., Luba, L. D., Cherry, J. A., & Maathuis, H. (1994). Field study of a long and very narrow contaminant plume. *Groundwater*, *32*(6), 1008–1016.
- Wada, Y., van Beek, L. P., van Kempen, C. M., Reckman, J. W., Vasak, S., & Bierkens, M. F.
 (2010). Global depletion of groundwater resources. *Geophysical Research Letters*, 37(20).

- Warr, O., Sherwood Lollar, B., Fellowes, J., Sutcliffe, C. N., McDermott, J. M., Holland, G., et al. (2018). Tracing ancient hydrogeological fracture network age and compartmentalisation using noble gases. *Geochimica et Cosmochimica Acta*, *222*, 340–362.
- Warr, O., Giunta, T., Onstott, T. C., Kieft, T. L., Harris, R. L., Nisson, D. M., & Sherwood Lollar, B. (2021). The role of low-temperature ¹⁸O exchange in the isotopic evolution of deep subsurface fluids. *Chemical Geology*, *561*, 120027.
- Wassmuth, F. R., Green, K., Arnold, W., & Cameron, N. (2009). Polymer flood application to improve heavy oil recovery at East Bodo. *Journal of Canadian Petroleum Technology*, *48*(02), 55–61.
- Winter, G. V., Bloomsburg, G., & Williams, R. E. (1983). Simulation of ground-water inflow into an underground lead mine. *Groundwater Monitoring & Remediation*, *3*(1), 65–69.
- Worth, K., White, D., Chalaturnyk, R., Sorensen, J., Hawkes, C., Rostron, B., et al. (2014). Aquistore project measurement, monitoring, and verification: from concept to CO2 injection. *Energy Procedia*, *63*, 3202–3208.
- Zalasiewicz, J., Williams, M., Fortey, R., Smith, A., Barry, T. L., Coe, A. L., et al. (2011). Stratigraphy of the Anthropocene. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, *369*(1938), 1036–1055.
- Zalasiewicz, J., Waters, C. N., & Williams, M. (2014). Human bioturbation, and the subterranean landscape of the Anthropocene. *Anthropocene*, *6*, 3–9.
- Zoback, M., & Smit, D. (2023). Meeting the challenges of large-scale carbon storage and hydrogen production. *Proceedings of the National Academy of Sciences*, *120*(11), e2202397120.

van der Zwaan, B., & Dalla Longa, F. (2019). Integrated assessment projections for global

geothermal energy use. *Geothermics*, 82, 203–211.



Figure 1: Approximate depths of subsurface activities. Median (31 m) and 95th (130 m) percentile of water wells (Jasechko & Perrone, 2021); minimum depth of CCS in sedimentary basins (800 m) (Benson & Cole, 2008); shallow limit of oil and gas development (including injection and disposal; 600 m) (Lemay, 2008); geothermal (>2,000 m) (Nardini, 2022). The upper temperature limit for life (80-121 °C) (Bar-On et al., 2018; Magnabosco et al., 2018)

approximately corresponds to the lowest temperatures required for geothermal power generation (Nardini, 2022; Tester et al., 2021). Circulation of meteoric water occurs up to depths of a few km (McIntosh & Ferguson, 2021) but fluxes are small below 500 m and residence times range from tens of thousands to millions of years (Ferguson et al., 2023; Jasechko et al., 2017; Warr et al., 2021).



Figure 2: Current oil and gas production involves similar fluid fluxes to natural deep (>500 m) groundwater flow with uncertainties based on 25th and 75 percentiles from a Cl flux-based estimate (Ferguson et al., 2023). Current geothermal projects are associated with smaller fluxes (C. E. Clark et al., 2010; IEA, 2021a). The range of projected fluxes for future CCS (Krevor et al., 2023; Zoback & Smit, 2023) and geothermal electricity production (van der Zwaan & Dalla

Longa, 2019) are similar to current fluxes from oil and gas production (IEA, 2021b). Scaling up Li extraction (Marza et al., 2023) from sedimentary basins to current global production from all sources considering 25th, 50th and 75th percentile of Li concentrations in produced waters would also require a similar amount of fluid.